Author's response to reviews

Title: Developing a decision aid to guide public sector health policy decisions: A study protocol

Authors:

Peggy Tso (Peggy.tso@utoronto.ca)
Anthony J Culyer (tony.culyer@utoronto.ca)
Melissa Brouwers (mbrouwer@mcmaster.ca)
Mark J Dobrow (mark.dobrow@utoronto.ca)

Version: 8 Date: 2 May 2011

Author's response to reviews: see over
May 2, 2011

Implementation Science Editorial Team

To whom it may concern,

In concordance with the suggested revisions made at the proofing stage, we have submitted a revised draft of our manuscript entitled ‘Developing a decision aid to guide public sector health policy decisions: A study protocol’. The following describes the changes that were made:

Major revisions (we require the author to make these changes)

1) Abstract: please provide a short Abstract of up to 350 words, giving a description of the findings of your article. This should be identical in the manuscript and the submission system

   An identical abstract as to the one included in the manuscript has been uploaded through the submission system.

2) References: Please make sure references are cited in numerical order. We notice reference 12 is not cited within the main text. Please make sure references continue to be cited in numerical order once this reference is cited.

   An incorrect citation was corrected in the text that correlates to reference 12; all references are placed in the correct order.

3) References: please provide a functioning link to website
   http://www.partnershipagainstcancer.ca/colorectalNet

   A working link has been inserted into the references section.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

1) Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

   The manuscript has been reviewed to remove any typographical errors.

We thank you for your assistance in improving this paper.

Yours sincerely,

Peggy Tso, BHSc, MSc Candidate
Department of Health Policy Management & Evaluation
University of Toronto

Anthony J Culyer, CBE, BA, Hon DEcon, Hon FRCP, FRSA, FMedSci
Professor
Department of Health Policy, Management and Evaluation
University of Toronto

Melissa Brouwers, BSc, MA, PhD
Associate Professor
Department of Oncology
McMaster University

Mark Dobrow, MSc, PhD
Assistant Professor
Department of Health Policy, Management and Evaluation
University of Toronto